Materials Today Physics 11 (2019) 100159

ELSEVIER

Contents lists available at ScienceDirect

**Materials Today Physics** 

journal homepage: https://www.journals.elsevier.com/ materials-today-physics

# Nanostructured potential well/barrier engineering for realizing unprecedentedly large thermoelectric power factors



N. Neophytou <sup>a, \*</sup>, S. Foster <sup>a</sup>, V. Vargiamidis <sup>a</sup>, G. Pennelli <sup>b</sup>, D. Narducci <sup>c</sup>

<sup>a</sup> School of Engineering, University of Warwick, Coventry, CV4 7AL, UK

<sup>b</sup> Department of Information Engineering, University of Pisa, Pisa, I-56122, Italy

<sup>c</sup> Department of Materials Science, University of Milano Bicocca, Milano, I-20125, Italy

#### ARTICLE INFO

Article history: Received 2 August 2019 Received in revised form 21 October 2019 Accepted 1 November 2019 Available online 11 November 2019

Keywords: Thermoelectricity Nanocrystalline materials Seebeck coefficient Energy filtering Modulation doping

#### ABSTRACT

This work describes, through the semi-classical Boltzmann transport theory and simulation, a novel nanostructured material design that can lead to unprecedentedly high thermoelectric power factors, with improvements of more than an order of magnitude compared to optimal bulk material power factors. The design is based on a specific grain/grain-boundary (potential well/barrier) engineering such that: i) carrier energy filtering is achieved using potential barriers, combined with ii) higher than usual doping operating conditions such that high carrier velocities and mean-free-paths are utilized, iii) minimal carrier energy relaxation is achieved after passing over the barriers to propagate the high Seebeck coefficient of the barriers into the potential wells, and, importantly, iv) an intermediate dopantfree (depleted) region is formed. Thus, the design consists of a 'three-region geometry', in which the high doping resides in the center/core of the potential well, with a dopant-depleted region separating the doped region from the potential barriers. It is shown that the filtering barriers are optimal when they mitigate the reduction in conductivity they introduce, and this can be done primarily when they are 'clean' from dopants during the process of filtering. The potential wells, on the other hand, are optimal when they mitigate the reduced Seebeck coefficient they introduce by: i) not allowing carrier energy relaxation, and ii) mitigating the reduction in mobility that the high concentration of dopant impurities causes. It is shown that dopant segregation, with 'clean' dopant-depletion regions around the potential barriers, serves this key purpose of improved mobility toward the phonon-limited mobility levels in the wells. Using quantum transport simulations based on the non-equilibrium Green's function method as well as semi-classical Monte Carlo simulations, we also verify the important ingredients and validate this 'clean-filtering' design.

© 2019 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### 1. Introduction

Thermoelectric (TE) materials have made dramatic progress over the last several years. The thermoelectric figure of merit *ZT*, which quantifies the ability of a material to convert heat into electricity, has more than doubled compared to traditional values of *ZT*~1, reaching values above *ZT*~2 in several instances across materials and temperature ranges [1-12]. The figure of merit is determined by  $ZT = \sigma S^2 T/(\kappa_e + \kappa_1)$ , where  $\sigma$  is the electrical conductivity, *S* is the Seebeck coefficient, *T* is the absolute temperature, and  $\kappa_e$  and  $\kappa_1$  are the electronic and lattice parts of the thermal

conductivity, respectively. The recent improvements in *ZT* are mostly attributed to drastic reductions of the lattice thermal conductivity in nanostructured materials and nanocomposites which has reached amorphous limit values at  $\kappa_1 = 1-2$  W/mK and below [1,3,13,14], as well as complex phonon dynamics materials [4]. With such low thermal conductivities, however, any further benefits to *ZT* must be achieved through the improvement of the thermoelectric power factor (PF)  $\sigma S^2$ , for which no noticeable progress has so far been achieved.

The lack of progress in the PF improvement is attributed to the adverse interdependence of the electrical conductivity and Seebeck coefficient via the carrier density, which proves very difficult to overcome. The most commonly explored direction in improving the PF is the 'conventional' energy filtering approach in nanocomposites and superlattices, in which built-in potential barriers

\* Corresponding author.

E-mail address: N.Neophytou@warwick.ac.uk (N. Neophytou).

https://doi.org/10.1016/j.mtphys.2019.100159

2542-5293/© 2019 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



block the cold low-energy carriers, while allowing the hot highenergy carriers to flow [15–30]. This energetic preference increases the Seebeck coefficient. On the other hand, this approach has still not been widely applied because the conductivity is also reduced in the presence of potential barriers. Current research efforts in improving the PF have thus diverted in many other directions. including: i) taking advantage of the density of states in lowdimensional materials through quantum confinement [31], or in bulk materials that include low-dimensional 'like' features [32,33], ii) band structure engineering and band-convergence strategies [32–36], iii) modulation doping and gating [37–45], iv) introducing resonances in the density of states [46], and even more recently v) concepts that take advantage of the Soret effect in hybrid porous/ electrolyte materials [47]. These approaches target improvements either in the Seebeck coefficient or in the electrical conductivity, with the hope that the other quantity will not be degraded significantly, and sometimes they report moderate PF improvements. However, no significant developments that lead to meaningful improvements, wider applicability, or generalization to many materials have been achieved by these methods either.

Recent efforts, both theoretical and experimental, by us and others, however, have revisited the energy-filtering concept and targeted designs that provide simultaneous improvements in both the electrical conductivity and the Seebeck coefficient to largely improve the PF  $\sigma S^2$ . Experimental works have indeed verified that it is possible to achieve very high PFs in nanostructured Si-based materials after undergoing specific treatment [48–52]. Measured data for PF improvements of over  $5 \times$  compared to bulk values were adequately explained using the Boltzmann transport theory [48,49]. In those cases, the grain boundaries of the nanocrystalline material serve as potential barriers resulting in energy filtering and high Seebeck coefficients. On the other hand, the carrier mean-freepath (MFP) for scattering (and mobility) in the grains was improved because of the fact that dopants were primarily placed in the middle regions of the grains, rather than being uniformly spread in the material—see later on Fig. 3a. The regions around the grain boundary potential barriers were depleted of dopants and allowed higher local carrier mobility and higher MFPs, which resulted in higher grain conductivities and overall material conductivities [48]. Thus, compared to other approaches, recent evidence suggests that energy filtering could be engineered in such a way as to lead to large PF improvements.

In light of the strong evidence of such exceptionally high PFs demonstrated, as well as the recent surge in efforts to use energy filtering and design the grain/grain-boundary system efficiently in a variety of materials [28-30,53], in this work, we re-examine theoretically energy filtering under degenerate conditions and dopant segregation. Using the Boltzmann transport theory, we provide an in-depth investigation of a generalized grain/grainboundary design concept and explore the ultimate upper limit that can potentially be achieved under realistic conditions. The design is based on a number of concepts/'ingredients', whose contributions are examined individually, and then all are combined until the final upper limit is reached. To further examine the validity of these 'ingredients', we also employ more sophisticated quantum transport simulations and Monte Carlo simulations. We show that PFs even beyond the exceptionally high value of PF ~30 mW/mK<sup>2</sup> can be achieved once a nanostructure is properly designed.

The paper is organized as follows: Section 2 describes the approach and the main features of the proposed design, based on what we refer to as '3-region, clean-filtering'. There, we validate the semi-classical transport model we use against p-type Si (the semiconductor material for which the largest improvements were experimentally achieved, but without loss in generality for the

material choice). Section 3 describes the PF performance of the proposed design, by exploring the improvements that the individual design 'ingredients' lead to as they are added to the design one by one. We then explore the PF improvements as the 'ingredients' are all utilized together, and their parameter values increased to very high, but still realistic levels, to provide an upper limit estimate for the PF. Section 4 then discusses the validity of the assumptions/'ingredients' using the non-equilibrium Green's function (NEGF) quantum transport method and Monte Carlo simulations. Finally, Section 5 concludes the work.

### 2. Approach

To model TE transport in the structure we examine, we start with the Boltzmann transport equation (BTE) formalism and calibrate our model to match the mobility of p-type bulk Si, so that we remain within realistic exploration boundaries. Within the BTE, the conductivity and Seebeck coefficient can be extracted as [54,55]:

$$\sigma = q_0^2 \int_{E_0}^{\infty} dE \left( -\frac{\partial f_0}{\partial E} \right) \Xi(E) \equiv \int_{E_0}^{\infty} \tilde{\sigma}(E) dE,$$
(1a)

$$S = \frac{q_0 k_B}{\sigma} \int_{E_0}^{\infty} dE \left( -\frac{\partial f_0}{\partial E} \right) \Xi(E) \left( \frac{E - E_F}{k_B T} \right) \equiv \frac{1}{\sigma} \int_{E_0}^{\infty} \tilde{S}(E) dE,$$
(1b)

where  $q_0$  is the electron charge and  $k_B$  is the Boltzmann constant. Note that for easiness, we define  $\tilde{\sigma}(E)$  and  $\tilde{S}(E)$  (here and throughout the text), as quantities which when integrated over energy give the conductivity and Seebeck coefficients  $\sigma$  and S (thus, they do not have the units of conductivity and Seebeck coefficient themselves). The transport distribution function  $\Xi(E)$  in Eq. (1a) is defined as:

$$\Xi(E) = \tau(E)v(E)^2g(E), \qquad (2a)$$

$$\tau(E) = \frac{\lambda_0 (E/k_B T)^r}{v(E)},\tag{2b}$$

where  $\tau(E)$  is the momentum relaxation time, v(E) is the band structure velocity, g(E) is the density of states, and  $\lambda_0$  is the MFP for scattering. In Eq. (2b), the energy dependence of the MFP is introduced with a characteristic exponent r that defines a specific scattering mechanism. In the case of phonon scattering, the MFP is energy independent, r = 0 (for 3D channels), and consequently the scattering rate is proportional to the density of states. Note that Eq. (2b) originates directly from the definition of the MFP, which is defined as the average time between momentum relaxing scattering events  $\tau(E) \times$  the carrier band structure velocity v(E) as  $\lambda(E) = v(E)\tau(E)$ . The MFP then can be expressed by a constant,  $\lambda_0$ , times an energy dependent term [56–58]. The constant can be adjusted to fit experimental low-field mobility data at low concentrations to mimic the transport properties of a specific material. In 3D structures, under isotropic scattering conditions (as for acoustic/optical phonons), the scattering rate is proportional to the density of final states which has an  $E^{1/2}$  trend. With the velocity following an  $E^{1/2}$  trend as well,  $\lambda(E) \sim E^{1/2} E^{-1/2} = E^0$ , which results in the MFP being energy independent.

For ionized impurity scattering (IIS), different expressions for the scattering rate and for the screening length apply at different doping concentrations [56]. Here we use the Brooks-Herring model with screening, which is satisfactory for doping concentrations up to  $10^{18}$  cm<sup>-3</sup> for mapping to the Si p-type mobility [55]. Above that

concentration (which is more relevant to our design), we merge to the strongly screened transition rate as described in Ref. [48], in an effort to match as close as possible the measured mobility of p-type Si [59–61]. Fig. 1a shows mobility calculations of p-type Si for the phonon-limited case (red line) and the phonon plus IIS limited case (blue line). By using  $\lambda_0 = 7.4$  nm for phonons, the desired bulk low-density mobility for p-type Si at 300 K is achieved ( $\mu = 450 \text{ cm}^2/\text{V-s}$ ). Our results agree particularly well at the carrier concentrations of interest, around  $p = 5 \times 10^{19} \text{ cm}^{-3}$ . The difference between the phonon-limited mobility and the phonon plus IIS mobility, which can be up to an order of magnitude, is a central aspect of our design as will be explained below.

With regards to thermoelectric performance, Fig. 1b–d show comparison of the phonon-limited to the phonon plus IIS conductivity, Seebeck coefficient, and thermoelectric PF, respectively, for p-type bulk Si where our model was calibrated on. The PF  $\sigma S^2$  peaks at high carrier concentrations around  $10^{19}-10^{20}/\text{cm}^3$ , and in most cases, this is achieved by doping, which introduces strong IIS and severely limits the carrier mobility and electronic conductivity. The PF would be at least a factor of >2× higher in the absence of IIS, which is the motivation behind modulation doping and gating methods [37–45]. To date, however, in such studies, the improvement over doped materials was only modest, in the best case the PF values were similar to those of the doped materials. We show below, however, how this can be utilized in a very efficient way.

Energy filtering through the introduction of potential barriers is the most commonly used approach to impede the flow of lowenergy carriers and increase the Seebeck coefficient. As indicated in Fig. 2a, electrons transport in the material through alternating potential wells and potential barriers. High-energy electrons gain energy to jump over the barriers and make it to the next well region. Once the electrons pass over the barrier and enter the potential well, there they tend to lose energy, usually through the emission of optical phonons (Fig. 2a), and relax around the Fermi level of the well within a few optical phonon emission MFPs,  $\lambda_E$ . Low-energy electrons are mostly blocked. In the following sections, we explain the model for electronic transport in the nanostructures



**Fig. 1.** (a) The mobility of p-type Si from experimental data (by Jacoboni [59]—black dots and Masetti [61]—green dots), the calculated phonon-limited mobility (red line), and the benchmarked phonon plus ionized impurity scattering from our semi-classical model (blue line). (b–d) The conductivity, Seebeck coefficient, and power factor for the cases of the phonon-limited (red lines) and phonons plus ionized impurity scattering (blue lines) considerations.

we consider, under different geometrical conditions, which turn out to influence the PF. We describe the equations that describe electronic transport in the potential wells and afterwards in the potential barriers, determining energy filtering.

#### 2.1. Well region—Filtering and electron flow in the potential well

When  $\lambda_{\rm E}$  is small ( $\lambda_{\rm E} \ll L_{\rm W}$ , where  $L_{\rm W}$  is the length of the well), transport in the barrier/well regions is essentially independent of each other (Fig. 2b). In this case, the total Seebeck coefficient and conductivity can be thought of as a simple combination of the individual quantities in each region, weighted by the size (volume fraction) of each region ('W'-well, 'B'-barrier) as [37]:

$$\frac{v_{\text{tot}}}{\sigma_{\text{tot}}} = \frac{v_{\text{W}}}{\sigma_{\text{W}}} + \frac{v_{\text{B}}}{\sigma_{\text{B}}}$$
(3a)

$$S_{\text{tot}} = \frac{S_{\text{W}}v_{\text{W}} + S_{\text{B}}v_{\text{B}}}{v_{\text{W}} + v_{\text{B}}} \tag{3b}$$

where, in 3D bulk materials, the weighting factor  $v_i$  is the volume of each region. The derivation of a generalized composite Seebeck coefficient equation as above can be found in the Appendix. The 'local' TE coefficients of the wells in this case are computed by simply considering the bulk value of these regions as:

$$\sigma_{\rm W} = \int_{0}^{\infty} \tilde{\sigma}_{\rm W}(E) dE, \tag{4a}$$

$$S_{\rm W} = \frac{1}{\sigma_{\rm W}} \int_{0}^{\infty} \tilde{S}_{\rm W}(E) dE, \tag{4b}$$

and those of the barriers by:

$$\sigma_{\rm B} = \int\limits_{V_{\rm D}} \tilde{\sigma}_{\rm B}(E) dE, \tag{4c}$$

$$S_{\rm B} = \frac{1}{\sigma_{\rm B}} \int_{V_{\rm B}}^{\infty} \tilde{S}_{\rm B}(E) dE.$$
(4d)

In this case, PF improvements cannot be easily achieved, as the improvement in the Seebeck coefficient is limited to the volume that the barriers occupy, and in those regions, the resistance is significantly increased (exponential drop in density).

#### 2.2. Well region—Energy non-relaxing case

In the opposite scenario with respect to carrier energy relaxation, i.e.,  $\lambda_E > L_W$ , electrons can flow over the barriers without (fully) relaxing their energy when they enter the wells (Fig. 2c). In this scenario, only carriers with energies above the barrier height  $V_B$  are contributing to transport. The Seebeck coefficient is benefitted in this case, as it is by definition the average energy of the current flow  $\langle E \rangle$  [26]:

$$S = \left(\frac{\langle E \rangle - E_F}{q_0 T}\right) \tag{5a}$$

The fact that the flow is high in energy throughout the channel (barriers and wells) essentially results in propagating the high Seebeck coefficient of the barrier into the well. In this energy non-relaxing case, the TE coefficients in the well region can then be computed by only considering transport above  $V_{\rm B}$  as:



electron flow: Thermionic - NON-momentum relaxing



**Fig. 2.** (a) Schematic of typical electron transport in a superlattice structure, where electrons flow over potential barriers and relax into potential wells. (b–c) Transport in a superlattice in the extreme cases where energy relaxation into the well after overpassing the barrier happens immediately and fully, and when energy relaxation does not happen at all. (d) Non-equilibrium Green's function simulations for the energy of the current flow in a superlattice (red line), indicating that carriers (blue line) relax into the wells partially, depending on the sizes of the well regions. The yellow/green colormap indicates the current flow I(E,x). (e) Schematics indicating situations resembling ballistic thermionic emission over the barrier (left), and emission over the barrier where carriers completely relax on the band edge at the top-of-the-barrier (right).

$$\sigma_{\rm W} = \int_{V_{\rm b}}^{\infty} \tilde{\sigma}_{\rm W}(E) dE \tag{6a}$$

$$S_{\rm W} = \frac{1}{\sigma_{\rm W}} \int_{V_{\rm B}}^{\infty} \tilde{S}_{\rm W}(E) dE.$$
 (6b)

Here, the only thing that changes from the treatment of each region independently is that the integral for the conductivity of the well begins from  $V_{\rm B}$ , rather than from the band edge of the well  $E_{\rm C} = 0$  eV.

In typical semiconductors,  $\lambda_E$  is in the order of several nanometers to a few tens of nanometers. Therefore, nanostructures in which the wells are a few tens of nanometers operate between the two cases of Fig. 2b and c. Fig. 2d shows a more realistic scenario of semi-relaxation of carriers. What is shown here is a quantum mechanical electronic transport simulation based on the NEGF method, including electron scattering with optical phonons in a superlattice-type structure. The dashed-blue line indicates the current flow (average of the yellow/green-flow colormap). The red-

solid line indicates the average energy of the current flow  $\langle E(x) \rangle$ , the quantity which when integrated and scaled over the channel provides the Seebeck coefficient. As observed, the current flows over the barriers by absorbing optical phonons and afterwards relaxes down in energy in the wells by emitting optical phonons. Because  $\lambda_{\rm E}$  in these simulations is set to  $\lambda_{\rm E} = 15$  nm and the length of the well to  $L_{\rm W} = 50$  nm, the current energy is not completely relaxed in the well; instead, this corresponds to the semi-relaxed case.

#### 2.3. Barrier region—Momentum relaxing case

Considering the design of the barrier regions now, the carriers can completely relax on the barriers (both their momentum and energy), in which case they are treated as independent regions from the wells. The right side of Fig. 2e shows the case where a carrier from the same state as before, 'relaxes' on the band structure of the barrier (blue line), then flows over the barrier, and 'relaxes' again on the 'local' band E(k) of the well. In this case, the barrier imposes a large resistance on the current flow and damages the overall conductivity of the material. The TE coefficients for the barrier can then be computed as in Eqs. (4c) and (4d).

# 2.4. Barrier region—Thermionic emission (momentum non-relaxing) case

However, if the barrier is narrow, carriers can be thermionically (ballistically) emitted over the potential barrier without relaxing on the top of it. The left side of Fig. 2e shows the case of thermionic emission, where the carriers from a given band structure state flow ballistically over the barrier and end in the same E(k) in the right side of the barrier (red arrow). We note that in reality for narrow barriers, the 'local' band structure could be different compared to the band structures of the constituent materials, and quantum mechanical reflections will appear because of a degree of well/ barrier state mismatch, which would add to interface resistance. These can be mitigated somewhat by employing smoothened barrier edges and oblique potentials (as in Fig. 3) rather than sharp ones [23,24], and we address this issue later on. The TE coefficients on the barrier region under thermionic emission conditions can then be computed at first order by allowing only the carriers from the wells with energies higher than the barrier heights  $V_{\rm B}$  over the barrier (ignoring contact resistance at this point) as:

$$\sigma_{\rm B} = \int_{V_{\rm B}}^{\infty} \tilde{\sigma}_{\rm W}(E) dE \tag{7a}$$

$$S_{\rm B} = \frac{1}{\sigma_{\rm B}} \int_{V_{\rm B}}^{\infty} \tilde{S}_{\rm W}(E) dE.$$
<sup>(7b)</sup>

#### 3. Nanostructured well/barrier design for very high PFs

#### 3.1. Designing optimal attributes for the filtering barrier

After describing the basic transport features and equations in SL structures, we begin our investigation by simulations of the optimized attributes for the potential barrier, and then for the potential well—targeting PF improvements. Fig. 4 shows row-wise the TE coefficients electrical conductivity  $\sigma$ , Seebeck coefficient *S*, and PF, respectively. Column-wise, it shows the change in these coefficients after each step of the design process we consider. We



**Fig. 3.** (a) The three-region structure design proposed in this work in a 2D top-down representation. The core of the well region (blue regions) is highly doped, the intermediate region between the core and the boundary (red region) is intrinsic (dopant-free), and the region in-between the wells is shown in grey. (b) A 'cut' through the dashed line of (a), indicating a simplification of the conduction band profile, with the doped regions, the intrinsic regions, and the barrier regions indicated. (c–d) Band profiles after self-consistent solutions of the Poisson equation for doping values  $N_D = 10^{19}/\text{cm}^3$  and  $5 \times 10^{20}/\text{cm}^3$ . In each sub-figure, cases for intrinsic regions of length 10 nm (red lines), 20 nm (blue lines), and 30 nm (black lines) are indicated (doped/undoped region boundaries are shown by the dashed lines). The Fermi level in (d) is denoted by the red-dashed line.

begin with the pristine channel, whose TE coefficients are shown by the black-dashed lines in all sub-figures for comparison, in which case the doping is uniformly distributed in the entire channel. The basic structure we consider has a well size  $L_W = 30$  nm, a barrier length  $L_B = 2$  nm, and barrier height  $V_B = 0.15$  eV (geometrical features which experimentally showed large PFs [48,49,51]).

As a first step, in the data of the first column of Fig. 4 (Fig. 4a–c), we introduce a potential barrier and consider independent transport in the well and barrier regions (i.e., full energy relaxation in the wells), as in the schematic of Fig. 2b. The TE coefficients for this system are given by Eq. (4). For the case where the entire structure is considered doped, the results shown by the green lines indicate strong reduction in  $\sigma$ , a slight increase in *S* as expected, but finally a strong reduction in the PF, at least for carrier concentrations up to ~10<sup>21</sup>.

In a second step, we consider the possibility that the potential barrier is free of dopants. Note that although the potential wells need to be doped to achieve the required carrier density, the barriers do not. In fact, other than them being formed because of bandedge discontinuities between the two materials (well and barriers), they can also be created by junctions of highly-doped/intrinsic regions, as shown in Fig. 3c and d. The rationale behind the undoped barrier regions originates from the much higher phonon-limited mobility and MFPs for scattering compared to those of the ionized impurity-limited transport (as shown in Fig. 1a). Here, the TE coefficients for the barrier can be computed as in Eqs. (4c) and (4d), but the scattering time is only determined by

electron—phonon scattering. In the case of an undoped barrier (Fig. 4a and c red lines),  $\sigma$  is degraded much less, *S* increases slightly, and the PF experiences a slight increase. Thus, as the barrier formation increases the material's resistance, by keeping it 'clean' from dopants, at least the MFPs are longer, and a portion of the conductivity is restored, mitigating the reduction in conductivity. This allows higher mobility carriers, limited locally by phonon scattering alone, rather than by the much stronger IIS (see Fig. 1a).

The third barrier attribute we investigate is the case where the carriers undergo thermionic emission above the barrier (blue lines), i.e., the case where the barrier size  $L_{\rm B}$  is much narrower compared to the momentum relaxation MFP of the carriers. Under the assumption of thermionic emission, the carriers from the wells that are not filtered and flow over the barriers, as indicated in the left of Fig. 2e, although subject to the doping in the wells, occupy high velocity states. In this case, similarly to the undoped barrier case, slight PF improvements are observed (blue line in Fig. 4c). The transport details over the barrier, however, are different in the two cases. In the undoped barrier case, the resistance is mitigated by the use of the higher mobility 'dopant-free' barrier region. In the thermionic case, the resistance is mitigated by the use of higher velocity carriers coming from the well, rather than the lower velocity carriers from the top of the barrier. The key outcome here is that 'clean' barrier regions or thermionic transport over them can restore the conductivity reduction that the potential barrier causes (for *reasonable V<sub>B</sub> values*). We now proceed with examining the optimal attributes of transport in the potential well.

# 3.2. Designing optimal attributes for the well—avoid energy relaxation

We next investigate electronic transport in the well region. At this stage, we consider that the well region is a uniformly, highly doped region. From here on, we assume the undoped barrier model, where charge carriers relax on the barriers. This is the conservative worst-case scenario compared to the thermionic emission assumption, but we also examine thermionic emission as an upper-limit scenario later on as well. As illustrated earlier in Fig. 2d, the current in a large portion of the well adjacent to the barriers propagates at larger energies (before it relaxes at lower energies), indicating both larger Seebeck coefficient and larger carrier velocities (and conductivity). In fact, it is these regions that provide PF improvements in superlattices and nanocomposites, because the high Seebeck coefficient of the barriers propagates into the wells, and high energies keep the conductivity still high. The optimal design is achieved when the carriers do not relax their energy into the well region after they overpass the barrier as in Fig. 2c. In practice, the barriers are placed at distances short enough to enforce less relaxation, but long enough to reduce the resistance introduced. Thus, an optimal compromise is achieved under semi-relaxation conditions. The advantage of energy semi-relaxation to the Seebeck coefficient is described by us and others in several works [26.27.62 - 64].

The TE coefficients in the limit of no-relaxation are shown by the red line in the second column of Fig. 4. in Fig. 4d-f. Essentially when the carriers flow at high energies in the wells, they raise the wells' low Seebeck coefficient to the high Seebeck values of the barriers (Fig. 4e), extending this high Seebeck in the entire channel. The downside is that only carriers with energies above  $V_{\rm B}$  are utilized, which would reduce the conductivity of the well and the overall channel conductivity. However, this is mitigated by the fact that the carriers of high energies are carriers of high mobility. The inset of Fig. 4d shows the mobility of p-type Si (dashed line) and the mobility of the carriers that travel above  $V_{\rm B}$ alone, which in this case is more than double (solid line). The conductivity is significantly lower compared to the pristine channel for lower densities (Fig. 4d), but as the Fermi level is raised to the V<sub>B</sub> level, the conductivity increases substantially. At those carrier densities, the Seebeck coefficient (Fig. 4e) is still high, and therefore the overall PF is improved compared to bulk by  $\sim 2 \times$  (Fig. 4f). This again illustrates the benefits of filtering under degenerate conditions [23,48].

Thus, to summarize the design of the filtering well/barrier system up to this point, one seeks for material designs which: i) allow for momentum non-relaxation on the top of the barrier, or 'clean' preferentially undoped barriers that restore/mitigate the conductivity reduction introduced by the potential barriers and ii) allow for energy non-relaxation in the well, which raises the low Seebeck coefficient of the wells to the high values of the barriers. Those criteria impose restrictions in the design, shape, and importantly the 'cleanliness' of the barrier (from dopants for example), to achieve large momentum relaxation lengths and potential well sizes comparable to  $\lambda_{\rm E}$  (or somewhat larger).

# 3.3. Filtering well/barrier optimization—a novel concept consisting of three regions

Moving forward, a novel filtering design geometry is introduced, and in the following sections, its performance is investigated. Although for the analysis we still employ the semiclassical Boltzmann transport formalism, later on quantum mechanical simulators and Monte Carlo simulators are also utilized to further validate some of the design 'ingredients'. A 2D top-down view of the proposed geometry is illustrated in Fig. 3a, where we now have rectangular domains of wells depicted by the blue-colored regions and barriers depicted by the greycolored regions. The wells can represent highly doped regions or regions where the band energy is generally lower (such that wells are formed).

The red-colored regions in between the heavily doped regions and the barrier regions are part of the wells (as in a nanocomposite material, for example), although for those regions we consider that the doping is reduced; for the purposes of this analysis, these regions are undoped. A simplified schematic of the potential profile in a 1D cross-section of the material is shown in Fig. 3b, with the middle doped regions being lowered in energy, the barriers regions' energy residing higher, and the potential of the middle region connecting the two extremes. Here we do not explore the details of the formation of the band profile in this region, but our goal is to illustrate the design principle.

In practice, the oblique potential profile in the middle region can be, for example, a result of the n++/i junction that is formed, pushing most of the depletion region in the undoped, intrinsic part. Its formation will be dictated by self-consistent electrostatics and can be extracted by solving the Poisson's equation together with carrier statistics as shown in Fig. 3c-d. In Fig. 3c, we simulate the potential profile of a double junction channel, consisting of regions doped at 10<sup>19</sup>/cm<sup>3</sup> at the left/ right sides, but left undoped in the middle. With the dashed lines, we show where the intrinsic region resides, having widths of  $L_i = 10$  nm (red lines), 20 nm (blue lines), and 30 nm (black lines). The solid lines show the self-consistently extracted potential profiles for each case, assuming p-type bulk Si density of states. Fig. 3d shows the same profiles, but in this case the doping in the left/right regions is raised to 5  $\times$   $10^{20}/$ cm<sup>3</sup> (the corresponding elevated Fermi level is depicted in Fig. 3d). Clearly, appropriate barriers for energy filtering are formed, with their shape and height controlled by the dopant values and intrinsic region length. In the rest of the paper, for simplicity, we assume that the potential profile in the undoped regions begins from the edge of the doped region and extends to the edge of the barrier region. (In principle, however, the potential barrier can be formed in a n++/i/n++ structure without the need for a different barrier material itself.) Because of the large differences in the doping, the depletion region indeed is shifted mostly in the undoped region, so it does effectively begin from the edge of the doped region. Note that the word 'depletion' throughout this work denotes both dopant and majority carrier depletion (compared to the n++ region) as in our previous works in Refs. [48,49]. In principle, the middleregion barrier can be optimized by varying the dopant distribution accordingly, but in this work, we assume intrinsic regions and provide the foundations of the design principle. We, thus, refer to these regions interchangeably as 'dopantdepleted', 'clean', or 'intrinsic'.

The TE coefficients in the case of the three-region structure can be computed by combining the individual coefficients of the three regions (well-W, intrinsic-i, and barrier-B) as:

$$\frac{v_{\text{tot}}}{\sigma_{\text{tot}}} = \frac{v_{\text{W}}}{\sigma_{\text{W}}} + \frac{v_{\text{i}}}{\sigma_{\text{i}}} + \frac{v_{\text{B}}}{\sigma_{\text{B}}},$$
(8a)

$$S_{\text{tot}} = \frac{S_{W}v_{W} + S_{i}v_{i} + S_{B}v_{B}}{v_{W} + v_{i} + v_{B}}.$$
(8b)



**Fig. 4.** The proposed design structure 'ingredients' one-by-one column-wise. Row-wise: the electrical conductivity, Seebeck coefficients, and power factors versus carrier density. In all cases, the black-dashed line shows the properties of the pristine flat band channel, without any barriers, included for comparison. Column 1 (a–c): Comparison between the pristine case, and superlattices where the well/barrier regions are both doped and independent of each other (green line), and cases where the barrier is undoped (red lines) and where thermionic emission prevails above the barrier (blue lines). The barrier height is  $V_B = 0.15$  eV. Column 2 (d–f): Comparison between the pristine case and superlattices when considering the undoped barrier case and completely unrelaxed current energies in the wells (red lines). Inset of d: The mobility of the isolated carriers that flow over the  $V_B$  (solid line), compared to the mobility of all carriers (dashed line). Column 3 (g–i): Comparison between the pristine case of undoped barriers, unrelaxed energies in the wells, bad opant-free regions of length  $W_i = 5$  nm in the well next to the barrier regions (blue lines). In dashed-red lines, the data from Column 3 (green lines). In dashed-red lines, the data from Column 3 (green lines). In dashed-red lines, the data from Column 3 are repeated for direct comparison.

In a manner similar to that of the wells and barrier regions, the conductivity and Seebeck coefficient of the dopant-depleted regions (labeled 'i' for intrinsic from here on), can be extracted by:

$$\sigma_{i} = \int_{V_{i}}^{\infty} \tilde{\sigma}_{i}(E) dE, \tag{9a}$$

 $\infty$ 

$$S_{i} = \frac{1}{\sigma_{i}} \int_{V}^{\infty} \tilde{S}_{i}(E) dE.$$
(9b)

In general, these quantities have a spatial dependence as the band edge changes in the 'i' region, but in most of the following analysis, we consider an average band contribution, with that band edge located at mid-energy  $V_B/2$  unless specified otherwise. (We have investigated various cases in our model, i.e., several band edge positions, each providing slightly different outcomes that do not change the foundation and advantages of our design.) We also note that the models described by Eqs. (1), (3), and (8) are strictly valid for 1D periodic systems. Nanocomposites, on the other hand, are described by a 3D aperiodic geometry, and the complexity of the transport paths is such that it would not allow us to map the 3D

onto 1D paths beyond a first-order estimation. The design we propose requires the current to flow normal to the wells and the barriers, rather than in parallel to them—i.e., the 2D geometry in Fig. 3a consists ideally of columnar grains extending into the page, as in Ref. [48]. Nanocomposites are also subject to geometry variations, and superlattice 1D geometries can also be considered as a limiting case for a nanocomposite system. Thus, we argue that Eqs. (1), (3) and (8) (with the volume fraction included rather than the length of the regions) are at first order applicable to nano-composite/nanocrystalline materials as well.

### 3.4. Thermoelectric coefficients in the structure with dopantdepleted regions

The TE coefficients for the '3-region' structure are shown by the blue lines in the third column of Fig. 4, Fig. 4g–i. Again, for comparison, we also show the pristine material properties by the black-dashed line and re-plot the result for non-relaxing wells of the second column by the red-dashed lines. The middle, dopant-depleted intrinsic region is assumed to be of width  $W_i = 5$  nm here, and the TE coefficients are computed using Eqs. (8) and (9). The middle region reduces the Seebeck coefficient compared to the non-relaxing wells of the second column (blue- vs. red-dashed

lines in Fig. 4h), but the dopant-free region strongly increases the conductivity of the overall domain (Fig. 4g). The carriers can now flow more easily under the weaker phonon-limited scattering conditions that prevail in the dopant-depleted regions. These regions geometrically occupy a significant volume of the structure even when having a narrow  $W_i = 5$  nm width. Thus, the overall conductivity acquires a significant phonon-limited part with longer MFPs, rather than an IIS limited part with much shorter MFPs. A significant PF improvement is then achieved in this case as shown by the blue line in Fig. 4i.

# 3.5. Designing optimal attributes for the three-region structure by allowing thermal conductivity variations

Finally, we add another component to the design of the material, which brings an independent improvement in the Seebeck coefficient without first-order changes in the conductivity. When different thermal conductivities ( $\kappa$ ) exist in the different regions, the overall Seebeck coefficient can be generalized to (see derivation in the Appendix) [27,37,64]:

$$S_{\text{tot}} = \frac{S_{\text{W}} \nu_{\text{W}} / \kappa_{\text{W}} + S_{i} \nu_{i} / \kappa_{i} + S_{\text{B}} \nu_{B} / \kappa_{\text{B}}}{\nu_{\text{W}} / \kappa_{\text{W}} + \nu_{i} / \kappa_{i} + \nu_{B} / \kappa_{\text{B}}}$$
(10)

Here, we assume a ratio of  $\kappa_W/\kappa_B = 5$  (and assume  $\kappa_i = \kappa_W$ ), which is a reasonable ratio between the conductivities of grains and grain boundaries, for example. An additional increase in the Seebeck coefficient is achieved as shown in Fig. 4k (green lines vs. bluedashed lines). This new component leads to a larger PF as shown in Fig. 4l. Here we included the TE coefficients from the second and third columns of Fig. 4 in dashed-lines for comparison. Overall, by considering these design strategies, a PF increase of ~5 × compared to the bulk material can be achieved. Note that such: i) dopantdepleted intrinsic regions, ii) semi-relaxation of energy, iii) undoped barriers, and iv) thermal conductivity variations were used to explain experiments that measured very high PFs even up to ~22 W/mK<sup>2</sup> [48,49,51]. In the following sections, we push the limits of what PFs can be achieved with this design principle.

### 3.6. Increasing the design parameters Wi, V<sub>B</sub>, and $\kappa_W/\kappa_B$

To demonstrate the full potential of this design, we now proceed by increasing at a higher degree (but still under practically achievable limits) the parameters that allowed for the PF improvements. We examine increases in *W*i, *V*<sub>B</sub>, and  $\kappa_W/\kappa_B$  one by one, and finally combine all three of them together. The calculated results are shown in Fig. 5, where row-wise, as earlier, we show the TE coefficients  $\sigma$ , S, and  $\sigma S^2$ , and column-wise the influence of increasing  $W_i$ ,  $V_B$ , and  $\kappa_W/\kappa_B$ , and all three simultaneously. In the sub-figures of the first three columns, we still show for comparison the lines from the previous investigation in Fig. 4l, which indicate the effect on the TE coefficients for each design ingredient (dashed lines). In the first column, by increasing  $W_i$  from 5 nm to  $W_i = 10 \text{ nm}$  (solid-blue lines), i.e., the length of the intrinsic regions is now longer, the conductivity increases significantly (Fig. 5a). Now a larger area of the material is composed of dopant clean regions, in which transport is phonon-limited with longer MFPs. The Seebeck coefficient is essentially not changed (Fig. 5b), despite the increase in the conductivity, because the elongated intrinsic regions raise the overall well band edge  $E_{\rm C}$  compared to the shorter  $W_{\rm i}$  case (see Fig. 3b–d). This means that the  $\eta_F = E_C - E_F$  is larger overall (in absolute terms), which tends to increase the Seebeck coefficient, thus mitigating the natural drop in the Seebeck coefficient when the conductivity is increased. Overall, therefore, the PF largely increases to values ~13 mW/mK<sup>2</sup>, dominated by the increase in the conductivity. Importantly, the PF maximum is achieved for slightly smaller densities, following the shift of the conductivity to lower densities, which is easier to achieve experimentally.

Next, we examine the increase in the barrier height  $V_B$  from  $V_B = 0.15$  eV to  $V_B = 0.25$  eV. The results are shown in the second column of Fig. 5 (Fig. 5d and f), by the red-solid lines. The conductivity is significantly reduced up to the high densities of  $10^{20}$ / cm<sup>3</sup>, where the Fermi level  $E_F$  has still not yet reached the  $V_B$ . At higher densities, when the  $E_F$  overpasses  $V_B$ , the conductivity is recovered to the pristine values, even slightly higher (with this slight increase depending on the details of the dopant-depleted regions  $W_i$  and how their consideration is accounted for). The Seebeck coefficient, as expected, is largely increased (Fig. 5e), which makes the PF to also largely increase (Fig. 5f). In this case, however, the PF peak is shifted toward the higher densities, again following the shift in the conductivity.

The next step is to examine the increase in the ratio  $\kappa_W/\kappa_B$  from 5 to 10 in the third column of Fig. 5, given by the green solid lines in Fig. 5g–i. Independently of the conductivity, the Seebeck coefficient is improved, which reflects on PF improvements in Fig. 5i.

Finally, in the fourth column of Fig. 5, magenta lines in sub-Fig. 5j–l, we consider these larger increases of all the parameters together, i.e., we increase the parameter values to  $W_i = 10$  nm,  $V_B = 0.25$  eV, and  $\kappa_W/\kappa_B = 10$ . Here we also kept the solid lines from each previous individual case for reference. In Fig. 5j, the conductivity is benefitted significantly from the presence of the dopantfree regions, and it is shifted to an intermediate region between the  $W_i = 10$  nm alone (blue line), and the  $V_B = 0.25$  eV alone (red line). Carriers still need high  $E_F$  levels to be able to overpass the increased  $V_B$ , which allows for large Seebeck improvements as seen in Fig. 5k. Putting it all together, the increase in the PF is quite substantial, reaching incredibly high values of >20 mW/mK<sup>2</sup> (magenta line in Fig. 5l), a factor of ~15 × over the bulk value where we started from (dashed black line).

These are some very high values predicted by our model and simulations. Our model is simple in considering transport and has also considered some idealized assumptions, such as energy non-relaxing transport and control on doping regions. However, we need to stress that even if those idealized conditions are not met in reality, there is still a lot of room for the PF to be improved substantially over the bulk values. In any case, an efficient TE material can still be realized with even  $5 \times$  improvement in the PF. It is also quite remarkable that we consider such high barriers  $V_B = 0.25$  eV, and the PF still peaks at densities around  $10^{20}$ /cm<sup>3</sup>, which are still achievable with current technologies for most TE materials. Thus, considering: i) this 'clean-filtering' approach, where the barrier is dopant-free, as well as ii) the support in the conductivity of the well from the dopant-depleted regions, a high conductivity is achieved at high Seebeck regions and incredibly high PFs can be realized.

# 3.7. Increasing $W_i$ , $V_B$ , and $\kappa_W/\kappa_B$ to the extreme for reaching the PF upper limit

Next, we go one step further and consider extreme conditions under which the three parameters  $W_i$ ,  $V_B$ , and  $\kappa_W/\kappa_B$  are increased to the very high values of  $W_i = 15$  nm,  $V_B = 0.3$  eV, and  $\kappa_W/\kappa_B = 15$ . In the extreme  $W_i = 15$  nm case, the entire well is dopant depleted (as  $L_W = 30$  nm), and it is assumed that all dopants reside in a deltafunction positioned in the middle of the well. In practice, a small heavily doped region can exist, which is there to supply the mobile carriers, without interfering with transport, an illustration of the modulation doping technique [37,40]. The TE coefficients of some combinations of these values are shown in Fig. 6a–c, indeed indicating incredibly high values of beyond 30 mW/mK<sup>2</sup> (the magentadashed line is the same as the magenta-solid line from Fig. 5 for comparison). It is quite instructive to show the mobility of these structures in Fig. 6d. The red-dashed line shows for reference the bulk phonon-limited mobility and the black-dashed line the bulk phonon plus IIS mobility, as in Fig. 1a. The proposed designs have much lower mobility for lower densities compared to both the phonon-limited and the phonon + IIS limited bulk mobilities. At higher densities, however, they all approach very closely the bulk phonon-limited mobility and significantly overpass the bulk IIS-limited mobility. Thus, the materials, at high densities, owing to the dominant presence of the large dopant-free regions, are overall phonon scattering limited.

### 3.8. Thermionic emission benefits

In the second part of Fig. 6 (Fig. 6e-h), we repeat the same calculations as in Fig. 6a–d, but in this case we consider thermionic emission over the barrier, as shown in the left panel of Fig. 2e and described by Eq. (7). The simple assumption of thermionic emission through a thin barrier allows the charge with energies higher than  $V_{\rm B}$  to flow 'freely' over the barriers and it is quite advantageous at higher  $V_{\rm B}$  (up to a limit), which otherwise introduces strong resistance to the current. Thus, in this case, the conductivities are much higher compared to the previous scenario (Fig. 6a vs. Fig. 6e), the mobilities are higher (Fig. 6d vs. Fig. 6h), which doubles the PF as well (Fig. 6c vs. Fig. 6g) to extremely high values of beyond 60 mW/mK<sup>2</sup>. Of course, this is an overestimated value, which will drop once we consider interface resistance or the resistance that arises through quantum mechanical well/barrier momentum state mismatch. This mismatch will be stronger with the barrier height as well. However, we examine the validity of this further below and point out that thin barriers where thermionic emission prevails can in general provide higher PFs.

We now devote the next part of the work to examining how realistic the conditions that we impose for obtaining such PFs are, using more advanced simulations. Specifically, we examine the position of the Fermi level at such high doping densities, the validity of the non- or semi-relaxation of current energy in our designs, the validity of thermionic emission together with the potential role of well/barrier state mismatch on the interface resistance, and the validity of the higher conductivity in SLs of n++/i junctions compared to uniformly doped channels. We also discuss the ideal volume fraction of the filtering barriers and potential practical implementations.

# 4. Discussion

# 4.1. Doping level and the Fermi level position

A major design ingredient in the design is the existence of a dopant-depletion region, whereas all the required doping resides in a small region in the middle of the grain. Thus, it is important to have an estimate where the  $E_{\rm F}$  resides with respect to the band edge  $E_{\rm C}$  in the doped region. We already expect to operate in the highly degenerate conditions (locally in the middle core of the wells, with the  $E_{\rm F}$  residing high into the bands), although in the barrier regions, of course, the  $E_{\rm F}$  will be below the band profile. In the dopantdepleted regions, the band will cross the  $E_{\rm F}$ . The question we essentially want to answer here is: how high can we dope the middle region, or how high will the  $E_{\rm F}$  be if a certain mobile carrier density needs to be achieved in the entire material region, given that it will only be supplied from the central/core region (bluecolored regions in Fig. 3a)? Ultimately, as shown in Fig. 3c and d, the built-in barriers are formed in the undoped region, which is largely extended because the depletion region is preferentially placed in the intrinsic region of the n++/i junction. On the other hand, the depletion region in the highly doped core is much narrower because of the very high doping (even just a few nanometers—see Fig. 3c and d for the non-flat band regions left/right of the dashed lines). Essentially, those narrow regions are depleted of the same amount of carriers that appear in the entire undoped barrier region,  $W_i + L_B$  combined. The illustrations of Fig. 3c and d are extracted from self-consistent Poisson equations, and they clearly indicate that just a few nanometers of highly doped regions are enough to support the depletion and barrier regions with the necessary carrier density needed for the PF to peak to very large values. The higher the doping of the highly doped regions, the smaller the extension of the depletion into the doped regions and the larger the depletion regions in the intrinsic/barrier regions (Fig. 3d).

In Fig. 7a, we show the PF computations for the case of the different depletion region sizes considered, i.e.,  $W_i = 5$  nm (blue line same as in Fig. 4i), 10 nm (red line), and 15 nm (entirely dopant depleted wells, with delta-function shaped core distributed doping, green line). Interestingly, the larger the dopant-depleted region, the higher the PF, but the smaller the required doping density in the middle of the well to achieve maximum PF, which would be easier to achieve experimentally anyway. Fig. 7b shows the position of the Fermi level in the middle region for the different cases of dopantfree region widths, in the structure where the barrier height is  $V_{\rm B} = 0.15$  eV as depicted. With the black-dashed line, we show the  $E_F$  position with respect to the band edge  $E_C = 0$  eV for the pristine uniformly doped channel case. Note that we consider p-type Si DOS parameters as high PFs are achieved for p-type Si [48], although we invert the bands to talk about conduction bands, as it is easier to perceive a flow over barriers than below barriers. As the regions where dopants are allowed (blue-colored regions in the middle of the wells in Fig. 3a) are reduced ( $W_i$  increases), a higher  $E_F$  positioning (higher center region doping) is needed to achieve the same carrier density. Around the density needed for maximum PFs  $5 \times 10^{19}$ /cm<sup>3</sup>, the  $E_{\rm F}$  is pushed into the band by 0.1 eV, almost  $3k_{\rm B}T$ higher than what would supply the same density if the structure was uniformly doped. The last case,  $W_i = 15$  nm, leaves no room for doping to be placed into, as the length of the well region we consider is  $L_W = 30$  nm. However, as it was stated above, it should be understood that there could be a narrow region of few nanometers of highly doped regions in excess, i.e., the well size would have been slightly larger. These are a small part of the well volume, still, however, enough to provide the necessary density in the intrinsic regions and the barrier regions.

The high doping in the central region then serves an important role, as it pushes the bands low for high velocities to participate in transport. The fact, however, that a narrow volume of the material is responsible for suppling the entire doping required (from  $10^{19}$  cm<sup>3</sup> $-10^{20}$ /cm<sup>3</sup>) requires that the E<sub>F</sub> levels increase substantially, to ~0.1 eV into the bands, with the doped regions reaching doping values at the levels of  $10^{20}$  cm<sup>3</sup>-5 ×  $10^{20}$ /cm<sup>3</sup> (to see this, one can find the  $E_{\rm F}$  in Fig. 7b corresponding to the maximum PF of Fig. 7a and then project horizontally that level until it meets the dashed line). Interestingly, from Fig. 7b we can extract that to achieve 5  $\times$  10<sup>19</sup>/cm<sup>3</sup> overall concentrations in the  $W_i$  = 15 nm structure, concentrations of only ~2  $\times$  10  $^{20}/\text{cm}^3$  in the middle bluecolored regions are needed. As indicated by the narrow depletion lengths of the heavily doped regions in Fig. 3c and d, a few nanometers of such central region would suffice. Such dopant values are realistic within current technologies, for example in Si, and even more importantly for nanocrystalline Si, where dopant solubility thresholds allow for concentrations as large as  $5 \times 10^{20}$  and  $2 \times 10^{21}$  cm<sup>-3</sup> using boron and phosphorus, respectively [65]. The simulations here are performed using Si parameters, but the concept we present is applicable in general to other materials as well.



**Fig. 5.** The proposed design structure 'ingredients' increased one-by-one column-wise to higher values. Row-wise: the electrical conductivity, Seebeck coefficients, and power factors versus carrier density. In all cases, the black-dashed line shows the properties of the pristine flat band channel, without any barriers included for comparison. The red-dashed, blue-dashed, and green-dashed lines are the corresponding solid line data from Fig. 4 for comparison. Column 1 (a–c): Increase in the dopant-free region from  $W_i = 5 \text{ nm to } W_i = 10 \text{ mm}$  (blue solid lines). Column 2 (d–f): Increase in the barrier rheight from  $V_B = 0.15 \text{ eV}$  to  $V_B = 0.25 \text{ eV}$  (red-solid lines). Column 3 (g–i): Increase in the ratio of the thermal conductivities of the well and barrier from  $\kappa_W/\kappa_B = 5 \text{ to } \kappa_W/\kappa_B = 10$  (green-solid lines). Column 4 (j–l): Increase in all the three parameters simultaneously to the levels of the data in Columns 1–3 (magenta-solid lines).

#### 4.2. Controlling the energy relaxation

An important aspect of the design is the reduced energyrelaxation in the well region, such that the high Seebeck coefficient of the barriers is transferred throughout the material. Thus, it is useful to point out what determines energy relaxation and what is the degree of energy relaxation under realistic conditions, as well as to propose some practical design directions, which provide control over relaxation. Regarding geometrical features for the size of the well, several works by us and others point out that energy relaxation is prevented, or mitigated, when the well sizes are in the range of the energy relaxation length,  $L_W \sim \lambda_E$  or smaller [26,27,64]. Energy relaxation in semiconductors is dominated by inelastic scattering processes, primarily electron-optical phonon scattering. In Si, for example, the electron–optical phonon MFP is around  $\lambda_{\rm E}$ ~15 nm, which results in well sizes of  $L_W$  ~50 nm to exhibit semirelaxation of the current energy [27,64]. Well sizes of the order of  $L_W = 30$  nm will only exhibit some degree of relaxation as we show below. However, other than the well size, there are other parameters that can contribute to reduced carrier energy relaxation in the design proposed, and they are discussed below.

Fig. 8a shows NEGF quantum transport simulations in SL channel geometries with semi-relaxing  $L_W = 50$  nm wells (Fig. 8a). For this, we have used our in-house 2D simulator that we have

developed with details explained in Refs. [26,64,66]. We include the effect of electron—phonon scattering with different common phonon energies from  $\hbar\omega$ = 0.02 eV to 0.09 eV (leading to different  $\lambda_E$ ), for an example case where  $V_B = E_F + 0.05$  eV. The colormap indicates the current flow in space and energy, with yellow indicating high current, whereas the curved lines indicate the energy of the current flow <*E*>, which determines the Seebeck coefficient as in Eq. (5). From our NEGF simulations, this is extracted by scaling the current  $I_{ch}(E,x)$  at all spatial points as:

$$\langle E(\mathbf{x})\rangle = \frac{\int_0^\infty I_{\rm ch}(E, \mathbf{x}) E dE}{\int_0^\infty I_{\rm ch}(E, \mathbf{x}) dE}.$$
(11)

Carriers flow over the barrier and then relax into the well in a distance determined by the relaxation length  $\lambda_E$ . In NEGF, we can control  $\lambda_E$  by adjusting the electron-phonon scattering strength,  $D_0$ , and the phonon energy,  $\hbar\omega$ . The short horizontal lines in the second well show the level where the  $\langle E \rangle$  of the current flow will reside at, in the case where the barriers are absent, i.e., the level at which the current energy relaxes in the pristine material. For all values of  $\hbar\omega = 0.02$  eV to 0.09 eV examined (energies for typical TE materials), the current is not relaxed at the pristine well level, but is



**Fig. 6.** The proposed design structure 'ingredients' increased one-by-one column-wise to extreme values of  $W_i = 15$  nm,  $V_B = 30$  eV, and  $\kappa_W/\kappa_B = 15$ . (a–d) Undoped barrier, independent well/barrier model. (a) The electrical conductivity. (b) The Seebeck coefficients. (c) The power factors. (d) The mobilities versus carrier density. In all cases, the black-dashed lines show the properties of the pristine flat band channel, without any barriers included and the magenta-dashed lines are the maximum power factor data from Fig. 5 for comparison. (e–h) The same quantities for the thermionic emission model.

higher, on average in between the energy levels observed on the top of the barriers and those in the pristine wells.

Indeed, in Fig. 8b we show the Seebeck coefficients of these channel materials with varying  $\hbar\omega$  (red line) and varying phonon strength  $D_0$  (blue line), but plotted as a function of the  $\lambda_E$  that they correspond to (again typical values for TE materials and semiconductors). The  $\lambda_E$  is extracted by fitting the relaxing  $\langle E \rangle$  by an exponential function  $e^{-x/\lambda_{\rm E}}$  for a structure with a single barrier as indicated in the inset of Fig. 8b. The dashed-flat lines show the Seebeck coefficients of the pristine channel  $S_{W}$ , and a channel consisting of one large barrier extending across its entire length, S<sub>B</sub>, for the  $\hbar \omega = 0.06 \text{ eV}$  (as in Si)—for clarity we do not show the lines for the uniform channels for the rest of the phonon energies, but we note that our calculations indicate a ~20% variation at most for the other phonon energies. The well size  $L_W \sim 50$  nm of this example dictates semi-relaxation for the Si case, and thus the overall SL Seebeck coefficient resides in the upper half of the allowed range between the pristine values  $S_B$  and  $S_W$  (despite the fact that the barriers occupy a much smaller portion of the channel compared to the wells). Materials with weak electron-optical phonon scattering, or materials with large optical phonon energies (right side of Fig. 8b), on the other hand, have weaker relaxation, with the Seebeck coefficient being even closer to that of the pristine barrier material S<sub>B</sub>.

Evidently, whether  $\hbar\omega$  or  $D_0$  is responsible in altering  $\lambda_E$ , the actual relaxation and overall Seebeck coefficients are different, even at the same  $\lambda_E$  (simply, the red/blue lines in Fig. 8b differ). The overall  $\lambda_E$  and Seebeck coefficient are mostly linear with electron—phonon coupling strength  $D_0$  (blue line). On the other hand, for very small, or very large,  $\hbar\omega$  values, even at the same expected  $\lambda_E$  as extracted from the single barrier cases, once a SL channel is formed, the actual relaxation is less, and the Seebeck coefficient is higher (higher red vs. blue line). The reasons are the following: *i*) *Small phonon energies:* When the phonon energy  $\hbar\omega$  is small, the  $\lambda_E$  decreases, but scattering approaches the elastic limit, and thus relaxation saturates, and is even suppressed for



**Fig. 7.** (a) The power factors of the structure with  $V_B = 0.15$  eV in the case of dopantdepleted regions of length  $W_i = 5$  nm, 10 nm, and 15 nm (the blue-solid line is repeated from Fig. 4i). (b) The position of the Fermi level  $E_F$  from the bottom of the well in each case versus carrier density. In both sub-figures, the black-dashed line shows the properties of the pristine flat band channel, without any barriers included for comparison. The presence of the dopant-free region forces the  $E_F$  higher compared to the pristine channel to achieve the same carrier density level.



**Fig. 8.** (a) The superlattice structure with the energy of the current flow  $\langle E \rangle$  for different optical phonon-energies 0.02 eV, 0.03 eV, 0.06 eV, and 0.09 eV. The short horizontal dashed lines indicate the energy of the current level in the pristine channel, i.e., where  $\langle E \rangle$  would relax to in an infinitely long well. The colormap indicates the current flow  $|(E_x)$ . (b) The Seebeck coefficient versus the energy relaxation length of a superlattice for the case where  $\lambda_E$  is altered by changing the phonon-energies (red line) and by altering the electron–phonon coupling strength (blue line). Inset: The extraction of  $\lambda_E$  by an exponential fit of  $\langle E(x) \rangle$  after the current passes over a single barrier. (c) The Seebeck coefficient of the superlattice structure versus the length of the well  $L_W$  (red line). The magenta line shows the Seebeck coefficient in the case of full and immediate relaxation after the carriers pass over the barriers. The dashed horizontal lines in (b) and (c) indicate the Seebeck coefficient of a pristine material without barriers. S<sub>W</sub> (infinite well) and of a pristine material with a large barrier S<sub>B</sub> (infinite barrier).

 $\hbar\omega \rightarrow 0$ . *ii*) Large phonon energies: For an electron at energy E to emit an optical phonon and relax at a lower energy, an available empty state has to exist at energy  $E - \hbar \omega$ . As electrons primarily flow over the barriers and into the wells, they would tend to relax around the Fermi level  $E_F$  (which in this case is placed into the bands at degenerate conditions). If, however,  $V_{\rm B}$  and  $E_{\rm F}$  are placed close in energy such that  $\hbar \omega > V_B - \max(E_F, E_C(x))$ , then energy relaxation is suppressed because of the absence of empty states below the Fermi level, also evident by the black-dashed line in Fig. 8a for the largest phonon energy we considered. Thus, the simplified picture of  $\lambda_{\rm E}$  extraction from a single barrier is no longer valid in SL structures, which tend to exhibit reduced energy relaxation, an advantageous observation. The point here, however, is that starting from the optimal conditions of a semirelaxed channel ( $L_W$  = 50 nm,  $\hbar \omega$  = 0.06 eV,  $\lambda_E$  ~ 15 nm), and by varying the phonon strength and the phonon energies such that  $\lambda_{\rm E}$  changes by 30% in either direction, still the Seebeck coefficient tends to move mostly toward that of the barrier, rather than that of the well in most of the parametric region.

Further on, to relate to the channels we have simulated earlier within the BTE and Si parameters, we have chosen a channel with  $\lambda_{\rm E} = 15$  nm and  $\hbar \omega = 0.06$  eV (resembling Si), and performed simulations for different well sizes from the ultra-short  $L_W = 2$  nm, up to  $L_W = 100$  nm. Fig. 8c shows the Seebeck coefficients for these structures (red line), whereas again the dashed-flat lines show the Seebeck coefficients of the pristine channel  $S_{W_{\rm c}}$  and the channel consisting of one large barrier  $S_{\rm B}$ . (The semi-relaxing  $L_W = 50$  nm channel we considered earlier has a Seebeck coefficient which is in the middle of the two limiting cases.) Clearly, however, for the  $L_W = 30$  nm channels that we considered in the BTE simulations earlier, the Seebeck coefficient is  $S_{SL}$  ~180  $\mu V/K$ , which is much closer to the  $S_B$ ~210  $\mu$ V/K, rather than the S<sub>W</sub> ~85  $\mu$ V/K. This justifies our considerations of non-relaxing current energy in the wells. (Note that in practice the optimal PF conditions are found when some relaxation is present, such that the barriers are spaced as far as possible to reduce the density of interface resistances, but as close as possible to prevent relaxation-we discuss interface resistance reduction directions further below.) Interestingly, the magenta-dashed line shows an analytical calculation of the SL Seebeck coefficient if the well/barrier is considered independently (as in Fig. 2b). In this case, much lower Seebeck coefficients are achieved for the SL.

# 4.3. Reduced relaxation and interface resistance in the three-region structures

It is also worth discussing a few other things that point toward reduced energy relaxation in the wells of the proposed design. For example, the shape of the band edge  $E_{\rm C}$  in the dopantfree regions, as shown in Fig. 3c and d, with oblique band edges: i) provides higher Seebeck coefficient because of the higher average  $E_{\rm C}$  and ii) further reduces the availability of empty states at lower energies for electrons to relax their energy into. This reduces the energy window for optical phonon emission to happen and thus, reduces the energy relaxation. This is clearly indicated in Fig. 9a, where we have performed NEGF simulations using electron-optical phonon scattering alone and altered the barrier potential to the oblique shape as a first-order approximation of what is shown in Fig. 3b. As the sidewalls of the barrier become more and more oblique, the energy in the wells begins to rise. The oblique sidewalls, however, tend to reduce the energy of the current flow above the barriers, but finally modest improvements to the overall Seebeck coefficient are observed (Fig. 9c). Furthermore, in general, highly doped regions (in the well core), also push the energy of the current upwards as lower energies scatter more effectively off dopants (evident from the anisotropic Brooks-Herring scattering model [56]). Thus, in the structures proposed, the combination of: i) the chosen length for the well to be in the order of  $\lambda_{\rm F}$ , ii) the 'oblique' band shape in the dopant-free regions, and iii) the doped core, allow for a significant degree of energy non-relaxing transport in the wells. The important point, here, is that the reduced energy relaxation in the wells is justified also by consideration of quantum transport simulations. This justifies the choice of the beneficial non-relaxing consideration in the BTE simulations earlier.

As a side note, we have shown in the past that nanoinclusions in the well regions, having barrier heights up to the  $V_B$  of the superlattice, can push the energy of the current flow upwards and further increase the Seebeck coefficient, also allowing for small, but noticeable PF improvements [26,66]. That would be something to also provide lower thermal conductivities, with additional benefits for *ZT*.



**Fig. 9.** (a) The energy dependence of the current flow (yellow-green colormap) in superlattice (SL) structures with oblique barrier sidewalls and the average energy of the current flow  $\langle E(x) \rangle$  (curved lines). The Fermi level is depicted by the flat blue line. The coloring of  $\langle E \rangle$  corresponds to the coloring of the barriers. (b–d) The conductance, Seebeck coefficient, and power factor of the SLs as a function of the sidewall inclination distance, *d*.

# 4.4. Oblique sidewalls reduce interface resistance

It is also quite interesting to observe the conductance of the channels in Fig. 9a, plotted against the sidewall distance *d*. The reduction in the 'local' Seebeck coefficient in the barriers in Fig. 9c at first when using oblique profiles is a signal of reduction in well/barrier interface resistance, a consequence of better well/barrier state matching. Indeed, our simulations show that the quantum mechanical transmission over the barriers increases in the oblique cases compared to the sharp barriers, and quantum reflections/ oscillations are smoothened out. Because of this, the conductance is improved at first instance by ~20% (Fig. 9b). For larger *d*, the conductance remains almost constant, whereas the Seebeck coefficient increases from contributions in the wells. Overall, the introduction of the sidewalls increases the PF monotonically, up to values of ~30% (Fig. 9d). Note that in these simulations the structures are undoped.

### 4.5. The validity of thermionic emission (Fig. 10)

When it comes to the behavior of carriers over the barrier, we have shown up to now that 'dopant-clean' barriers, with oblique sidewalls for reduced interface resistance, and/or thermionic emission from the wells over the barriers are important ingredients for the design. Here, we examine the validity of the thermionic emission, again using NEGF simulations, this time including electron-acoustic phonon scattering only (elastic scattering to isolate the effect of carrier relaxation on the barrier from inelastic relaxation processes into the wells). We simulate a channel with a single potential barrier in the middle and vary the length of the barrier  $L_B$  from  $L_B = 100$  nm (taking over the entire channel) to  $L_{\rm B} = 5$  nm and then to zero, i.e., the pristine channel case (as illustrated in the inset of Fig. 10). In Fig. 10, we plot the energy resolved transmission function, defined as  $Tr = (h/q_0^2)I_{\rm ch}/(f_1 - f_2)$ , where  $I_{ch}$  is the NEGF-extracted current, h is Plank's constant, and  $f_1$ and  $f_2$  are the Fermi–Dirac distributions of the left/right contacts, respectively [26]. All these quantities are energy dependent. The Tr is directly related to the transport distribution function in the BTE and has a linear dependence in energy in the case of acoustic phonon scattering for a single sub-band [66,67]. That linear dependence is captured in the (multi-band) NEGF simulations for



**Fig. 10.** Indication of the degree of relaxation of carriers on the barrier material as they propagate over it. The figure shows the NEGF-calculated energy resolved transmission function *T*<sup>r</sup> of the carriers in a channel with a single potential barrier with length  $L_B$  as indicated in the inset. Electron scattering with acoustic phonons only is considered in the calculation. Cases for different barrier lengths are shown from a large  $L_B = 100$  nm taking over most of the channel (black-dashed line) to a pristine channel (brown line). The 'jump' of the *Tr* from that of the larger barrier to that of the pristine material would indicate that carriers are more thermionically emitted over the barrier rather than relaxing on it as the barrier length is scaled.

the pristine channel (brown line), as well as the long barrier channel (purple-dashed line), with the initial point being the band edge, i.e. 0 eV in the pristine case, and  $V_{\rm B} = 0.05$  eV in the long barrier case.

As the barrier length  $L_{\rm B}$  is scaled, however, there is a clear shift at energies after V<sub>B</sub> toward the Tr of the pristine channel. The 'jump' in the Tr after  $V_B$  in the shorter  $L_B = 5$  nm barrier channel clearly indicates that carriers 'see' the barrier, but for energies above the barrier they have a Tr more similar to that of the well. This would be an indication of ballistic thermionic emission, in which case the carriers do not relax (at a large degree) on the bands of the barrier, i.e., overall they do not acquire the low velocities at the top of the barrier, but propagate with the well attributes. The gradual change of the black line Tr in Fig. 10 toward that of the well (reaching at ~80% of that value within 0.01 eV after the barrier at  $V_{\rm B} = 0.05$  eV) could signal that some well/barrier state mismatch and quantum reflections are still present, adding to the interface resistance. We also note that the aforementioned energy window deviations from the pristine Tr would increase as  $V_{\rm B}$  increases because of larger mismatch. However, any acquired slope in Tr being larger compared to the one of the large non-thermionic well (brown line in Fig. 10) would be beneficial to the conductivity and the PF. This degrating effect would be reduced, however, when: i) the barrier sidewalls acquire a slope as explained above in the discussion of Fig. 9a and ii) if the barrier material has much more transport modes compared to the well material, such that more momentum state matching is achieved [68]. This can be the case of a barrier material with much higher effective mass, for example. Thus, the optimal PF of our design could be somewhere in between the  $\sim$ 30 mW/mK<sup>2</sup> and  $\sim$ 60 mW/mK<sup>2</sup> indicated in the two examples of Fig. 6.

#### 4.6. High electronic conductivity in n++/i SL structures (Fig. 11)

Because the major new element this work proposes is the introduction of the dopant-free region separating the highly doped core or the potential well and the barrier, Monte Carlo simulations were further performed to verify the outcomes of the simple model employed in the main part of this paper. We employ a home-



**Fig. 11.** Schematics of structures simulated within Monte Carlo. (a) Pristine, uniformly doped channel (the dots indicate the doping placement). (b) Uniformly doped superlattice material. (c) Superlattice material consisting of a series of doped/intrinsic regions (n++/i). The position of the Fermi level is indicated. The barrier shape is extracted from a self-consistent solution of the Poisson equation (and used in (b) as well). (d) The energy resolved current in the three channels, in black for the pristine channel, in red for the uniformly doped SL, and in blue for the n++/i superlattice.

developed 2D electron transport Monte Carlo code, and as a simple example we simulate three channels as indicated in Fig. 11a-c: i) a uniformly doped channel without barriers, ii) a uniformly doped channel, but with barriers, and iii) a channel where only the well regions are doped, and the barriers are not. The shape of the barriers is extracted from a self-consistent Poisson solver, corresponding to doped regions of  $L_{doped} = 30$  nm at  $N_D = 4 \times 10^{19}/$  $cm^3$  and  $L_{undoped} = 20$  nm. P-type Si density of states were used. It is not our intention here to perform a complete study using the numerically expensive Monte Carlo simulations, but to indicate that indeed doping variation can mitigate the conductivity reduction due to the introduction of potential barriers. Thus, considering the simplicity of non-relaxing transport, we only consider elastic scattering (acoustic deformation potential scattering and IIS). We report on the electronic conductances of these three structures. i) The uniformly doped channel has  $G = 9.36 \times 10^{-6}$  S. ii) The conductance of the uniformly doped channel, but with barriers as well, suffers a large reduction down to  $G = 1.78 \times 10^{-6}$  S, a factor of  $5.3 \times$ . iii) 'Cleaning-up' the dopants from the barrier regions (as is the actual case for the n++/i self-consistent simulation here) helps the conductance to increase to  $G = 5.92 \times 10^{-6}$  S, recovering to ~60% of that of the uniform channel. The behavior is clearly observed in the energy resolved current of electrons in the Monte Carlo simulator, shown in Fig. 11d for the three cases. The current in the uniformly doped SL (red line) case picks up after the barrier, but even then remains lower compared to the pristine channel case (black line), as carriers on the band edge on the top of the barrier have lower velocities. On the other hand, the third case of the nonuniformly doped channel (blue line) picks also after the barrier, but it is significantly higher compared to the pristine case, contributing to increased conductivity. Finally, as the introduction of the barriers (by simple considerations) increases the average energy of the current  $\langle E \rangle$  by 2.5×, a more than doubling of the PF would be expected. The closest simulations that this can relate to in the earlier BTE simulations is what is observed in Fig. 4i at the data points of the corresponding densities.

# 4.7. Intrinsic Seebeck coefficient of the boundaries and their volume fraction

The boundary regions (grey regions in Fig. 3a) are there to facilitate the creation of the dopant-depleted regions (as for example in the nanocrystalline structures of Ref. [48-51]). The boundary regions are also important in providing the barrier  $V_{\rm B}$ , which will form when the boundary material and the potential well material have a band edge discontinuity,  $\Delta E_{C}$ . On the other hand, even if the boundary material does not have a significant  $\Delta E_{\rm C}$  (or even if it has a negative one), the presence of the junction between the heavily doped well core and the undoped/lightly-doped middle and boundary regions will electrostatically form an effective barrier  $V_{\rm B}$ . Thus, the isolated 'intrinsic' Seebeck coefficient of the boundary material will only have a secondary effect. The primary determination of the Seebeck coefficient and PF will be determined by the effective built-in barrier (band edge discontinuity and electrostatics). Note that in the case where the junctions (blue/red regions in Fig. 3a) are achieved with selective local doping on monocrystalline materials, even in the absence of boundary regions large benefits will also be expected. Where boundaries, as a second phase, can provide an additional improvement to the overall Seebeck coefficient, is through variability in the thermal conductivity.

With regards to the volume fraction that the boundaries occupy, within the assumption that they remain undoped, our simulations show that moderate PF benefits are possible (even up to  $2\times$ ) compared to the uniformly doped, pristine material, even if the boundaries occupy a larger volume fraction compared to the wells/ grains. To obtain the very large PF improvements we present. however, the boundary regions need to be smaller than 5 nm in length, to mitigate electrical resistance and allow for a degree of thermionic emission (as in Fig. 10). On the other hand, boundary regions smaller than 3 nm will allow quantum mechanical tunneling and result in smaller temperature drops across them, both of which reduce the Seebeck coefficient [24]. Thus, we suggest that the boundaries are optimally of the order of 5 nm thickness. In the case of Si, for example, to achieve semi-relaxation of the current flow in the channel, the grain/well size should be in the order of  $L_W = 30 \text{ nm} - 50 \text{ nm}$ . This leads to an optimal volume fraction ratio of at most ~25% for the boundary regions.

#### 4.8. Example of possible practical realization

The simplest experiment to design and evaluate the potential of the 'clean-filtering' approach is to begin with two regions and fabricate 2D superlattices formed of n++/i or n++/n-junctions. In that case, the barriers are formed in the intrinsic or lightly doped regions, which will be regions 'cleaner' of dopants, having phonon-limited mobility. Lithography can be used for the definition of windows through an oxide layer (grown by thermal oxidation, for example) on an silicon-on-insulator (SOI) wafer, shaped by lithography and etching to act as a mask for the doping process. Oxide windows, and hence the final doping concentration, can be arranged to form a 2D array using ion implantation (for example). As a next step, one can go even further by lithographically defining lines in the x- and y-directions to form a square 'net', and then dopant diffusion can create highly doped islands in the regions between the 'square net' lines as in the blue-colored islands of Fig. 3a.

### 5. Conclusions

In conclusion, this work proposes a novel design direction, which will allow nanostructured materials to deliver exceptionally high thermoelectric PFs, even more than an order of magnitude compared to the original material's corresponding values. The design is based on an extension and generalization of previously a presented strategy that realized experimentally very large PFs ( $5 \times$ compared to optimized pristine material values) [48,49]. In this work, it is shown that much higher PFs can be achieved once the grain/grain-boundary (well/barrier) design is properly optimized. Specifically, the proposed design utilizes energy filtering where carriers flow from heavily doped potential wells into undoped barriers (with a degree of thermionic emission) for reducing the barrier resistance that ionized impurities would have caused. Importantly, though, it introduces an undoped region, 'clean' of dopant impurities, that separates the core of the wells from the barriers. This essentially allows for higher carrier MFPs and mobility in the wells (approaching phonon-limited), compensating by far the conductivity reduction caused by the barriers. The work also points out that other than the three-regions (1-heavily doped well core, 2-intrinsic carrier path spacer, and 3-potential barrier), an essential ingredient is that the energy of the carriers does not relax significantly in the well regions. It is shown, however, that this is the most probable and realistic case in the design we propose, as: i) the potential well length can be chosen such as to be in the same order as the energy relaxation MFP, ii) the oblique band edge shape of the intrinsic regions favors reduced relaxation, and iii) degenerate doping conditions also favor reduced energy relaxation. Thus, the design can provide exceptionally high PFs because it: i) reduces the resistance of the barriers, ii) transfers the high Seebeck coefficient of the barriers into the wells, and iii) allows very high conductivity in the wells. The latter is achieved by utilizing dopant-free intrinsic regions for transport, but with high-energy carriers that provide high carrier MFPs and phonon-limited rather than IIS dominated, mobilities. Although for some of the parameters, the simulations employed are relevant to Si, the design approach can be applied in general to other materials as well.

#### **Declaration of competing interest**

We declare no conflicts of interest related to this work.

### Acknowledgments

This work has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Programme (Grant Agreement No. 678763). We acknowledge Hans Kosina and Mischa Thesberg for helpful discussions.

# Appendix

Here we provide the theoretical proof of the equation for the Seebeck coefficient of a composite system, which leads to Eq. (3b) in the main paper. The Seebeck coefficient of an arbitrary irregular system is defined as the weighted average of the local Seebeck coefficients along the length of the material (*L*), with the weighting factor being the lattice temperature gradient as:

$$S = \frac{\int_{0}^{L} S(x) (dT_{\rm L}/dx) dx}{\Delta T}$$
(A1)

where for the applied temperature difference  $\Delta T$ , it holds  $\Delta T = \int_{-\infty}^{\infty} (dT_L/dx)$ . Here we assume a 1D channel material. We assume that the lattice temperature  $(T_L)$  varies according to a simple thermal circuit model. In this case, each of the two materials forming the composite system (barriers and wells) has a different temperature gradient across it depending on its thermal conductivity, as  $dT_L/dx|_B$  for the barriers and  $dT_L/dx|_W$  for the wells. The entire temperature drop across the material is then decomposed as:

$$\Delta T = \frac{dT_{\rm L}}{dx} \Big|_{\rm B} L_{\rm B} + \frac{dT_{\rm L}}{dx} \Big|_{\rm W} L_{\rm W} \tag{A2}$$

where  $L_{\rm B}$  and  $L_{\rm W}$  are the total lengths of the wells and barrier regions. At an interface between different materials, the heat flux is conserved, thus using Fourier's law we have:

$$J_{\rm Q} = -\kappa_{\rm B} \frac{dT_{\rm L}}{dx}\Big|_{\rm B} = -\kappa_{\rm W} \frac{dT_{\rm L}}{dx}\Big|_{\rm W}$$
(A3)

Substituting (A3) into (A1), we reach:

$$S = \frac{S_B L_B \frac{J_Q}{\kappa_B} + S_W L_W \frac{J_Q}{\kappa_W}}{L_B \frac{J_Q}{\kappa_B} + L_W \frac{J_Q}{\kappa_W}} = \frac{\frac{S_B L_B}{\kappa_B} + \frac{S_W L_W}{\kappa_W}}{\frac{L_B}{\kappa_B} + \frac{L_W}{\kappa_W}}$$
(A4)

In the case of a 3D material with irregularities in the distribution of the wells/barriers, for example as in a polycrystalline material of grains/grain boundaries, the lengths are replaced by the volume fractions of the different regions [37], i.e., the overall Seebeck coefficient of a 3D nanocomposite material is approximately the volume weighted average of the Seebeck coefficients of the constituent material phases as:

$$S = \frac{\frac{S_{B}\nu_{B}}{\kappa_{B}} + \frac{S_{W}\nu_{W}}{\kappa_{W}}}{\frac{\nu_{B}}{\mu_{B}} + \frac{\nu_{W}}{\kappa_{W}}}$$
(A5)

#### References

- [1] D. Beretta, et al., Mater. Sci. Eng. R Rep. 138 (2019) 210-255, https://doi.org/ 10.1016/j.mser.2018.09.001.
- [2] H.J. Wu, L.-D. Zhao, F.S. Zheng, D. Wu, Y.L. Pei, X. Tong, M.G. Kanatzidis, J.Q. He, Nat. Commun. 5 (2014) 5515.
- [3] K. Biswas, J. He, I.D. Blum, C.-I. Wu, T.P. Hogan, D.N. Seidman, V.P. Dravid, M.G. Kanatzidis, Nature 489 (2012) 414-418.
- [4] L.-D. Zhao, S.-H. Lo, Y. Zhang, H. Sun, G. Tan, C. Uher, C. Wolverton, V.P. Dravid, M.G. Kanatzidis, Nature 508 (2014) 373-377. [5] K.F. Hsu, et al., Science 80 303 (5659) (2004) 818-821.
- T. Fu, et al., J. Mater. 2 (2) (2016) 141-149.
- [7] B. Poudel, Q. Hao, Y. Ma, Y. Lan, A. Minnich, B. Yu, X. Yan, D. Wang, A. Muto, D. Vashaee, X. Chen, J. Liu, M.S. Dresselhaus, G. Chen, Z. Ren, Science 320 (2008) 634.
- W. Xie, X. Tang, Y. Yan, Q. Zhang, T.M. Tritt, Appl. Phys. Lett. 94 (10) (2009).
- [9] H. Liu, et al., Nat. Mater. 11 (5) (2012) 422-425.
- [10] S.I. Kim, et al., Science, 80 348 (6230) (2015) 109-114.
- [11] A.A. Olvera, et al., Energy Environ. Sci. 10 (7) (2017) 1668-1676.
- [12] G. Rogl, et al., Acta Mater. 63 (2014) 30-43.
- [13] Y. Nakamura, M. Isogawa, T. Ueda, S. Yamasaka, H. Matsui, J. Kikkawa,
- S. Ikeuchi, T. Oyake, T. Hori, J. Shiomi, A. Sakai, Nano Energy 12 (2015) 845. [14] J.A.P. Taborda, M.M. Rojo, J. Maiz, N. Neophytou, M.M. González, Sci. Rep. 6 (2016) 32778.
- [15] D. Vashaee, A. Shakouri, Phys. Rev. Lett. 92 (2004): 106103.
- [16] J.-H. Bahk, Z. Bian, A. Shakouri, Phys. Rev. B 89 (2014): 075204.
- [17] A. Popescu, L.M. Woods, J. Martin, G.S. Nolas, Phys. Rev. B 79 (2009) 205302.
- [18] C.J. Vineis, A. Shakouri, A. Majumdar, M.G. Kanatzidis, Adv. Mater. 22 (2010) 3970.
- [19] A. Shakouri, Annu. Rev. Mater. Res. 41 (2011) 399-431.
- [20] J.-H. Bahk, A. Shakouri, Phys. Rev. B 93 (2016) 165209.
- [21] L.-D. Zhao, V.P. Dravid, M.G. Kanatzidis, Energy Environ. Sci. 7 (2014) 251 - 268.
- [22] M. Zebarjadi, K. Esfarjani, M.S. Dresselhaus, Z.F. Ren, G. Chen, Energy Environ. Sci. 5 (2012) 5147-5162.
- [23] Neophytos Neophytou, Hans Kosina, J. Appl. Phys. 114 (2013): 044315.
- [24] M. Thesberg, M. Pourfath, H. Kosina, N. Neophytou, J. Appl. Phys. 118 (2015) 224301
- [25] M. Thesberg, M. Pourfath, N. Neophytou, H. Kosina, J. Electron. Mater. 45 (2016) 1584–1588.
- [26] Vassilios Vargiamidis, Neophytos Neophytou, Phys. Rev. B 99 (2019): 045405.
- [27] R. Kim, M.S. Lundstrom, J. Appl. Phys. 111 (2012): 024508.
- [28] C. Gayner, Y. Amouyal, Energy filtering of charge carriers, 1901789, Adv. Funct. Mater. (2019).
- [29] S. Sakane, et al., Materials Today Energy 13 (2019) 56-63.
- [30] T. Ishibe, et al., ACS Appl. Mater. Interfaces 10 (2018) 37709-37716.
- [31] L.D. Hicks, M.S. Dresselhaus, Phys. Rev. B 47 (1993) 16631.

- [32] A. Zevalkink, D.M. Smiadak, J.L. Blackburn, A.J. Ferguson, M.L. Chabinyc, O. Delaire, J. Wang, K. Kovnir, J. Martin, L.T. Schelhas, T.D. Sparks, S.D. Kang, M.T. Dylla, G.J. Snyder, B.R. Ortiz, E.S. Toberer, Appl. Phys. Rev. 5 (2) (2018): 021303.
- [33] Y. Pei, X. Shi, A. LaLonde, H. Wang, L. Chen, G.J. Snyder, Nature 473 (2011) 66-69.
- [34] Y. Tang, Z.M. Gibbs, L.A. Agapito, G. Li, H.-S. Kim, M.B. Nardelli, S. Curtarolo, G.J. Snyder, Nat. Mater. 14 (2015) 1223.
- [35] P. Norouzzadeh, D. Vashaee, Sci. Rep. 6 (2016) 22724.
- [36] C. Kumarasinghe, N. Neophytou, Phys. Rev. B 99 (2019) 195202.
- [37] M. Zebarjadi, G. Joshi, G. Zhu, B. Yu, A. Minnich, Y. Lan, X. Wang, M. Dresselhaus, Z. Ren, G. Chen, Nano Lett. 11 (2011) 2225–2230.
- [38] A. Samarelli, L. Ferre Llin, S. Cecchi, J. Frigerio, T. Etzelstorfer, E. Müller, J. Sallaferin, L. Ferre Lin, S. Ceccin, J. Frigeris, T. Eccestorier, L. Hauer, Y. Zhang, J.R. Watling, D. Chrastina, G. Isella, J. Stangl, J.P. Hague, J.M.R. Weaver, P. Dobson, D.J. Paul, J. Appl. Phys. 113 (2013) 233704.
- [39] Y.-L. Pei, H. Wu, D. Wu, F. Zheng, J. He, J. Am. Chem. Soc. 136 (2014): 13902.
- [40] N. Neophytou, M. Thesberg, J. Comput. Electron. 15 (1) (2016) 16–26.
  [41] Q.R. Hou, B.F. Gu, Y.B. Chen, Y.J. He, J.L. Sun, Appl. Phys. A 114 (2014) 943–949.
- [42] B.M. Curtin, E.A. Codecido, S. Krämer, J.E. Bowers, Nano Lett. 13 (2013) 5503-5508.
- [43] N. Neophytou, H. Kosina, Appl. Phys. Lett. 105 (2014): 073119.
- [44] J. Moon, J.-H. Kim, Z.C.Y. Chen, J. Xiang, R. Chen, Nano Lett. 13 (2013) 1196 - 1202.
- [45] B. Yu, et al., Nano Lett. 12 (4) (2012) 2077–2082.
- [46] C.M. Jaworski, V. Kulbachinskii, J.P. Heremans, Phys. Rev. B 80 (2009) 125208. [47] L. Márquez-García, B. Beltrán-Pitarch, D. Powell, G. Min, J. García-Cañadas, ACS Appl. Energy Mater. 1 (2018) 254-259.
- [48] N. Neophytou, X. Zianni, H. Kosina, S. Frabboni, B. Lorenzi, D. Narducci, Nanotechnology 24 (2013) 205402.
- [49] N. Neophytou, X. Zianni, H. Kosina, S. Frabboni, B. Lorenzi, D. Narducci, J. Electron. Mater. 43 (6) (2014) 1896–1904.
- [50] D. Narducci, B. Lorenzi, X. Zianni, N. Neophytou, S. Frabboni, G.C. Gazzadi, A. Roncaglia, F. Suriano, Phys. Status Solidi A 211 (6) (2014) 12551258.
- [51] B. Lorenzi, D. Narducci, R. Tonini, S. Frabboni, G.C. Gazzadi, G. Ottaviani, N. Neophytou, X. Zianni, J. Electron. Mater. 43 (10) (2014) 3812-3816.
- [52] Nick S. Bennett, Daragh Byrne, Aidan Cowley, Neophytos Neophytou, Appl. Phys. Lett. 109 (2016) 173905.
- [53] Kuo, et al., Grain boundary dominated charge transport in Mg3Sb2-based compounds, Energy Environ. Sci. 11 (2018) 429.
- [54] G.D. Mahan, J.O. Sofo, The best thermoelectric, Proc. Natl. Acad. Sci. U.S.A. 93 (1996) 7436-7439.
- [55] Neophytos Neophytou, Hans Kosina, Phys. Rev. B 83 (2011) 245305.
- [56] M. Lundstrom, Fundamentals of Carrier Transport, Cambridge University
- Press, 2000. [57] C. Jeong, R. Kim, M. Luisier, S. Datta, M. Lundstrom, J. Appl. Phys. 107 (2010): 023707.
- [58] J. Mao, J. Shuai, S. Song, Y. Wu, R. Dally, J. Zhou, Z. Liu, J. Sun, Q. Zhang, C. dela Cruz, S. Wilson, Y. Pei, D.J. Singh, G. Chen, C.-W. Chu, Z. Ren, Proc. Natl. Acad. Sci. 114 (40) (2017) 10548.
- [59] C. Jacoboni, L. Reggiani, Rev. Mod. Phys. 55 (1983) 645.
- [60] http://www.ioffe.ru/SVA/, ('Physical Properties of Semiconductors').
- [61] G. Masetti, M. Severi, S. Solmi, IEEE Trans. Electron Devices 30 (1983) 764.
- [62] J.Y.W. Seto, J. Appl. Phys. 46 (1975) 5247.
- [63] J.W. Orton, M.J. Powell, Rep. Prog. Phys. 43 (1980) 1263.
- M. Thesberg, H. Kosina, N. Neophytou, J. Appl. Phys. 120 (2016) 234302. [64]
- N.H. Nickel, P. Lengsfeld, I. Sieber, Phys. Rev. B 61 (2000) 15558. [65]
  - [66] S. Foster, M. Thesberg, N. Neophytou, Phys. Rev. B 96 (2017): 195425.
  - [67] N. Neophytou, H. Kosina, Phys. Rev. B 84 (2011): 085313.
  - [68] R. Kim, C. Jeong, M. Lundstrom, J. Appl. Phys. 107 (2010): 054502.